Author's response to reviews

Title: Does school-based physical activity decrease overweight and obesity in children aged 6-9 years? A two-year longitudinal study in the Czech Republic

Authors:

Erik Sigmund (erik.sigmund@upol.cz)
Walid El Ansari (walidansari@glos.ac.uk)
Dagmar Sigmundová (dagmar.sigmundova@upol.cz)

Version: 2 Date: 5 April 2012

Author's response to reviews:

- addition of Informed parental consent in first part of Methods: 'Written informed consent was obtained from parents of all children participating in the study'.
- addition of part 'Competing interests' after the Conclusions: 'The authors declare that they have no competing interests'.
